Author's response to reviews

Title: The independent effects of maternal obesity and gestational diabetes on the pregnancy outcomes

Authors:

Hayfaa A Wahabi (umlena@yahoo.com)
Amel A Fayed (fayeda@ksu-hs.edu.sa)
Rasmieh A Alzeidan (ras_zeidan@hotmail.com)

Version: 3 Date: 9 June 2014

Author's response to reviews:

Dear Editor,

We thank the reviewer for his valuable comments and below is our reply to his comments:

1. We agree with the reviewer that there are differences between the HAPO study and our study in the characteristics of the cohort therefore we removed that reference from the discussion and replace it with another reference relevant to our study population and supportive of our findings (highlighted in yellow).